RCF status

Eric Lancon
elancon@bnl.gov

Smooth operation
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Data archived last week
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90 1B last week
2 PB since 2/05

180 TB last week
6.3 PB since 2/05

NoO Issue
with HPSS



Data transters last week
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@ Total Inbound Current: 195.61 Average: 1.78 G Maximum: 8.92 G
W Total Outbound Current: 16.34 K Average: 77.43 M Maximum: 401.98 M

No backlog, transfers to ~nominal rates end of week
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Star Switch as0515 - 806/807 Channel 17/18 Aggregate - Traffic
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@ Total Inbound Current: 11.32 G Average: 3.87 G Maximum: 12.45 G
W Total Outbound Current: 60.98 M Average: 19.90 M Maximum: 65.65 M

May 17, 2016



